Author's response to reviews

Title: Lack of association between PRNP 1368 polymorphism and Alzheimer's disease or vascular dementia

Authors:

Byung-Hoon Jeong (bhjeong@hallym.ac.kr)
Kyung-Hee Lee (puremind72@empal.com)
Yun-Jung Lee (jiung0301@hallym.ac.kr)
Yun-Joong Kim (yunkim@hallym.ac.kr)
Young-Hoon Kim (sml-yh@smlab.co.kr)
Young-Sook Cho (yscho@smlab.co.kr)
Richard I. Carp (Richard.carp@omr.state.ny.us)
Yong-Sun Kim (yskim@hallym.ac.kr)
Eun-Kyoung Choi (ekchoi@hallym.ac.kr)

Version: 4 Date: 1 April 2009

Author's response to reviews: see over
Re: MS 1466152002197806, Formatting Changes

Dear The BioMed Central Editorial Production Team

Thank you for your letter of March 27th, 2009, implying that our manuscript entitled “Lack of association between PRNP 1368 polymorphism and Alzheimer’s disease or vascular dementia” by Jeong et al. will be published in the BMC Medical Genetics pending formatting requirements. We have now carefully revised the manuscript according to the comments raised by the The BioMed Central Editorial Production Team.

The following formatting changes were made:

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Response: According to the editor’s suggestion, we checked the author list and changed “Yun-Joong Kim” to “Yun Joong Kim”. “Eun-Kyoung Choi” should be co-author because of contributions in submission and revision states of this manuscript.

Table size - These must fit into an A4 portrait page. If you cannot re-orientate the table onto a single a4 portrait page (maximum size 176mm x 230mm), please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.

Response: According to the editor’s suggestion, we changed three tables to a single A4 portrait page.

Table order - The first table referenced in the manuscript text should be labelled Table 1. The second, Table 2, etc. The tables should be provided in this same order at the end of the manuscript file. Every table provided at the end of the manuscript file must be
Response: The first table referenced in our manuscript text has been already labeled as Table 1 in Methods section [Page 6, line 1]. The second and third tables referenced in the text have also been labeled as Table 2 and Table 3 in the Results section.

We hope these changes are to your satisfaction and that our revised manuscript meets with your final requirement.

Very sincerely yours,

Yong-Sun Kim, M.D., Ph.D.

Ilsong Institute of Life Science, Hallym University, Ilsong Building,
1605-4 Gwanyang-dong, Dongan-gu, Anyang,
Gyeonggi-do 431-060, South Korea
TEL : 82-31-380-1986, FAX : 82-31-388-3427
E-mail : yskim@hallym.ac.kr